Author’s response to reviews

Title: Optimising decision making on illness absenteeism due to fever and common infections within childcare centres: development of a multicomponent intervention and study protocol of a cluster randomised controlled trial

Authors:

Kirsten Peetoom (kirsten.peetoom@maastrichtuniversity.nl)
Rik Crutzen (rik.crutzen@maastrichtuniversity.nl)
Jolijn Bohnen (m.bohnen@student.maastrichtuniversity.nl)
Rachèl Verhoeven (rachel.verhoeven@maastrichtuniversity.nl)
Marjorie Nelissen-Vrancken (m.nelissen@medicijngebruik.nl)
Geert-jan Dinant (geertjan.dinant@maastrichtuniversity.nl)
Jochen Cals (j.cals@maastrichtuniversity.nl)

Version: 2 Date: 12 Jul 2017

Author’s response to reviews:

Dear editor,

According to the request sent on Jul 10th we amended the text regarding "consent for publication" section to "not applicable" (p14).

Our manuscript will not contain data from any individual person (details/images/videos). Primary and secondary outcomes will only be reported on group level with no reference to individual persons.

Participants provided informed consent to participate in this study.
Yours sincerely,

Kirsten Peetoom

Maastricht University